Author’s response to reviews

Title: Out-of-pocket expenditure by the senior citizens with chronic disease: the effect of total number of chronic diseases, specific chronic disease, and comorbid clusters

Authors:

M. Mofizul Islam (mofizul.islam@anu.edu.au)
Laurann Yen (Laurann.Yen@anu.edu.au)
Jose M Valderas (J.M.Valderas@exeter.ac.uk)
Ian S McRae (ian.s.mcrae@anu.edu.au)

Version: 2 Date: 1 July 2014

Author’s response to reviews:

Dear Rita,

As per your instruction I have now (i) included email addresses of all the authors in the title page, and (ii) formatted the section heading of the abstract both in the manuscript and submission system, structured into separate sections: Background, Methods, Results, and Conclusions.

Kindly do not hesitate to ask me of any further information or editing if you think necessary.

Thanking you for considering our manuscript for BMC Public Health,

Sincerely yours,

M. Mofizul Islam